Author’s response to reviews

Title: Professional centred shared decision making: patient decision aids in practice in primary care

Authors:

Duika Burges Watson (duika.burges-watson@ncl.ac.uk),
Richard G Thomson (richard.thomson@ncl.ac.uk),
Madeleine J Murtagh (m.j.murtagh@newcastle.ac.uk)

Version: 8 Date: 4 January 2008

Author’s response to reviews:

Removed the words 'Submitted solely to BMC Health Services Research' from the title page.

Provided different email address for each author, on the title page, beneath the corresponding author details.

Removed keywords.

Removed highlighting from manuscript text.

Revised sentence case for all headings and sub-headings in the manuscript, including table headings.

Moved the tables to a position after the references in a section entitled 'Tables'.

Authors contributions -> Authors’ contributions

Moved competing interests section to a position before the Authors' contributions.

Place Appendix before the Acknowledgements section of the manuscript.